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Abstract 

Castor oil has the potential to be a renewable resource for the synthesis of high-value-added 

chemicals. Currently, the first-generation castor oil derivatives such as sulfonated castor oil-

turkey red oil, ethoxylated castor oil, and hydrogenated castor oil are produced in India. 

These second and third generation derivatives has more values than the first-generation 

products. The technological challenges are higher temperature for conversion, development 

of suitable catalyst, and green process for production. The thesis focuses on conversion of 

castor oil derivatives from methyl ricinoleate to sebacic acid, 2-octanol, heptaldehyde and 

methyl undecenaote by using thermal-catalytic cracking, and alkali-splitting reaction. 

Transesterification of castor oil to convert into methyl ricinoleate for the physico-chemical 

characteristics and fatty acid composition of methyl ricinoleate and castor oil have been 

studied by conventional and modern techniques. It was observed that castor oil contained 

86.4 % ricinoleic acid in its fatty acid composition.  

This work focuses on the thermal cracking of methyl ricinoleate, the major component of 

castor oil, in a micro fixed bed reactor that was used to produce methyl undecenoate and 

heptaldehyde. Methyl undecenoate and heptaldehyde have wide applications in polymers, 

cosmetics, and drug industries. The effect of flow rate, sweeping gas flow rate, preheating, 

and reaction temperatures were examined briefly. Thermal cracking of methyl ricinoleate at a 

preheating temperature of 350 °C and reaction temperatures of 550 °C, the highest yields of 

methyl ricinoleate, 46.7 wt.%, and heptaldehyde 27.3 wt.% were achieved. The increase in 

sweep flow rate from 15 ml min-1 to 25 ml min-1 enhanced the production of pyrolytic oil 

yield to 92.6 wt.%. Density functional theory (DFT) was performed to examine the 

experimental results and distribution of the products involved in the dissociation of methyl 

ricinoleate. Assuming C-C bond scission followed by the cleavage of the OH bond, the 

reaction energy of methyl undecenoate and heptaldehyde thermal cracking is computed using 
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DFT. This research may provide an alternative pathway to produce important chemicals from 

castor oil for upscaling and process development for the industry.  

Furthermore, methyl undecenoate and heptaldehyde formation are enhanced in the presence 

of a catalyst. It has been demonstrated that lanthanum supported on zeolite HZSM-5 is 

effective at converting methyl ricinolate to bio-oil. The synthesis of lanthanum-loaded on 

HZSM-5 has been further confirmed through the characterizations of the catalyst. The 

experimental results indicate that the optimum condition for achieving maximum yield 

conversion was attained at 500 °C for 150 mins with 2 gm of 5 wt.% La/HZSM-5. The bio-

oil obtained at the optimal condition was comprised around 97.5 wt.% with methyl 

undecenoate (76.5 wt.%) and heptaldehyde (54.7 wt.%) alongside other minor aromatic 

hydrocarbons.  

Methyl ricinoleate was used as the raw material for alkali splitting to prepare sebacic acid and 

2-octanol. The reaction parameters, including catalyst, the ratio of oleochemicals/NaOH, 

reaction time, and reaction temperature, were optimized. It was found that 10wt.% Fe/HZSM-

5 showed the best catalytic performance, and 280 oC was considered the most suitable 

reaction temperature. The oleochemicals/NaOH ratios 10:15, 12:15, 14:15, and 15:15 were 

determined as the optimal ratio for alkali-splitting of methyl ricinoleate. In addition, the 

optimal reaction time of alkali-splitting of methyl ricinoleate was 4 hr, and the maximum 

yield was 5 wt.% Fe/HZSM-5 of 29.7 wt.% of 2-octanol and 56.8 wt.% of sebacic acid, 10 

wt.% Fe/HZSM-5 of 39.4 wt.% of 2-octanol and 68.1wt.% of sebacic acid, 15 wt.% 

Fe/HZSM-5 of 42.3 wt.% of 2-octanol and 71.8 wt.% of sebacic acid, respectively. FT-IR 

and NMR confirmed high purity of sebacic acid and 2-octanol.  

This research may provide an alternative pathway to produce important chemicals from 

castor oil for upscaling and process development for the industry. 
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साराााांश 

 

इस थीससस में वसणित शोध कार्ि में समथाइल रिसकनोसलएट की दो सबसे महत्वपूणि प्रसतसिर्ाएं शासमल 

हैं। हेप्टाल्डिहाइड, समथाइल अनडेसीनोएट, सेबाससक एससड औि 2-ऑके्टनॉल, औद्योसिक रूप से 

महत्वपूणि उच्च मूल्यवान ओलेओकेसमकल्स की असधकतम पैदावाि प्राप्त किने के सलए थमिल औि 

कैटेसलसटक िैसकंि औि क्षाि सवभाजन। 

समथाइल रिससनोलेट औि अिंडी के तेल की भौसतक-िासार्सनक सवशेषताओ ंऔि फैटी एससड संिचना 

के सलए अिंडी के तेल को समथाइल रिससनोलेट में परिवसतित किने के सलए अिंडी के तेल का 

ट्ांसएस्टिीसफकेशन पािंपरिक औि आधुसनक तकनीको ंद्वािा अध्यर्न सकर्ा िर्ा है। र्ह देखा िर्ा सक 

अिंडी के तेल में फैटी एससड संिचना में 86.4% रिससनोसलक एससड होता है। 

अिंडी के तेल में उच्च मूल्यवसधित िसार्नो ं के संशे्लषण के सलए एक नवीकिणीर् संसाधन होने की 

क्षमता है। र्ह कार्ि समथाइल रिससनोलेट के थमिल िैसकंि पि कें सित है, जो माइिो सफक्स्ड बेड 

रिएक्टि में अिंडी के तेल का प्रमुख घटक है सजसका उपर्ोि समथाइल अनडेसीनोएट औि 

हेप्टाल्डिहाइड का उत्पादन किने के सलए सकर्ा जाता था। समथाइल अनडेसीनोएट औि हेप्टाल्डिहाइड 

का पॉसलमि, सौदंर्ि प्रसाधन औि दवा उद्योिो ंमें व्यापक अनुप्रर्ोि है। प्रवाह दि, व्यापक िैस प्रवाह 

दि, प्रीहीसटंि औि प्रसतसिर्ा तापमान के प्रभाव की संके्षप में जांच की िई। 350 सडग्री सेल्डल्सर्स के 

प्रीहीसटंि तापमान औि 550 सडग्री सेल्डल्सर्स के प्रसतसिर्ा तापमान पि समथाइल रिसकनोलेट की थमिल 

िैसकंि, समथाइल रिसकनोलेट 46.7 wt.% औि हेप्टाल्डिहाइड 27.3 wt.% की उच्चतम उपज प्राप्त की 

िई। स्वीप प्रवाह दि को 15 समली समनट-1 से बढाकि 25 समली समनट-1 किने से पार्िोलाइसटक तेल 

उपज का उत्पादन 92.6 डबू्ल्यटी% तक बढ िर्ा। समथाइल रिसकनोलेट के पृथक्किण में शासमल 

उत्पादो ं के प्रर्ोिात्मक परिणामो ं औि सवतिण की जांच किने के सलए घनत्व कार्ाित्मक ससद्ांत 
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(डीएफटी) का प्रदशिन सकर्ा िर्ा था। सी-सी बांड सवखंडन को ओएच बांड के दिाि के बाद मानते हुए, 

समथाइल अनडेसीनोएट औि हेप्टाल्डिहाइड थमिल िैसकंि की प्रसतसिर्ा ऊजाि की िणना डीएफटी का 

उपर्ोि किके की जाती है। र्ह शोध उद्योि के सलए अपसे्कसलंि औि प्रसिर्ा सवकास के सलए अिंडी 

के तेल से महत्वपूणि िसार्नो ंका उत्पादन किने के सलए एक वैकल्डिक मािि प्रदान कि सकता है। 

सेबाससक एससड औि 2-ऑके्टनॉल तैर्ाि किने के सलए क्षाि सवभाजन के सलए कचे्च माल के रूप में 

समथाइल रिसकनोसलएट का उपर्ोि सकर्ा जाता था। उते्प्रिक, ओलेओकेसमकल्स/NaOH का अनुपात, 

प्रसतसिर्ा समर् औि प्रसतसिर्ा तापमान ससहत प्रसतसिर्ा मापदंडो ंको अनुकूसलत सकर्ा िर्ा था। र्ह 

पार्ा िर्ा सक 10wt.% Fe/HZSM-5 ने सबसे अच्छा उते्प्रिक प्रदशिन सदखार्ा, औि 280 oC को सबसे 

उपरु्क्त प्रसतसिर्ा तापमान माना िर्ा। ओलेओकेसमकल्स/NaOH अनुपात 10:15, 12:15, 14:15, 

15:15 को समथाइल रिसकनोलेट के क्षाि-सवभाजन के सलए इष्टतम अनुपात के रूप में सनधािरित सकर्ा 

िर्ा था। इसके अलावा, समथाइल रिससनोलेट के क्षाि-सवभाजन का इष्टतम प्रसतसिर्ा समर् 4 घंटे था, 

औि 5 wt.% Fe/HZSM-5 की असधकतम उपज, 29.7 wt.% 2-ऑके्टनॉल औि 56.8 wt.% सेबाससक 

एससड, 10 wt.% Fe/HZSM-5, 39.4 wt.% 2-ऑके्टनॉल औि 68.1 wt.% सेबाससक एससड, 15 wt.% 

Fe/HZSM-5, 42.3 wt.% 2-ऑके्टनॉल औि 71.8 wt.% िमशः  सेबाससक एससड का। सेबाससक एससड 

औि 2-ऑके्टनॉल की उच्च शुद्ता की पुसष्ट एफटी-आईआि औि एनएमआि द्वािा की िई थी। 
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